Author's response to reviews

Title: Do changes in social and economic factors lead to changes in drinking behavior in young adults?

Authors:

Frederieke S van der Deen (frederieke.vanderdeen@otago.ac.nz)
Kristie N Carter (kristie.carter@otago.ac.nz)
Sarah K Mckenzie (sarah.mckenzie@otago.ac.nz)
Tony Blakely (tony.blakely@otago.ac.nz)

Version: 2  Date: 20 May 2014

Author's response to reviews:

Dear the editor,

Following recommendations, line and page numbers are now inserted. We also provided text on the ethics approval and author’s contributions.

kind regards,
Frederieke Sanne van der Deen